LHE®E T RKFEFER
36 HBo5H J. University of Shanghai for Science and Technology Vol.36 No.5 2014

XEHS 1007 —6735(2014) 05— 0502 — 05 DOI:10.13255/j. enki. jusst.2014.05.018

SEMEMRSHSERTR NO HXRHF R

X 7%‘7’51 . ¥ 75(5%:2 s g‘k‘%“%ﬁz ,  BAE %2
Q. ;g AR 2E B ST i 5 40 4 TR BE . BifE 201418;
2. RigRSE R MU S 3 TRE2#BE, 1ifg 200240)

WE: RAEADRANFRER EEXNAFHRES L ERX AW REAAY ZAEGES BB EY
AR FRER NO #AFTEBAR. EREAN . L —ZHEMRARAE, 2 RaENRTALLR
NOSRFHEAERERAZ;NOZLRAFRMBEI H MG AAARENBEELIGZEHLEBR
SERBE,FRABAFET ;B3 b B AMRAL L5 5T AL R NO B LA
PLOASKRIT Bmed R RER. EALEBEE N, & F b AN R AER NO K FE & &L 80%, %
iE A W AR FRALA) ) 69 — FRARAT 69 7 X

KW AW Bab; R NO; FRAA A
FERES: TK6 XErFRERD: A

Experimental Study on NO Reduction by Reburning of
Biomass Gasification Syngas with Tar

LIU Chun-yuan', LUO Yong-hao*, ZHANG Rui-zhi*,  YIN Ren-hac®
(1. College of Urban Construction and Safety Engineering , Shanghai Institute of Technology , Shanghai 201418 , China ;
2. School of Mechanical and Power Engineering , Shanghai Jiaotong University , Shanghai 200240 , China)

Abstract: The technology of biomass gasification for reburning was adopted and the experimental
research on NO reduction by reburning of biomass gasification syngas with tar, produced from an
updraft biomass gasifier,was carried out on a tubular flow reactor. The experimental results show
that there is optimum oxygen content under a certain syngas flow rate. NO reduction efficiency
increases with the rise of temperature, but soot generated leads to decline of NO reduction
efficiency when oxygen is not enough under higher temperature. The effect of tar on NO reduction
can be fully reflected by comparing the NO reduction tendency by using syngas with tar and light
gases respectively. NO can be reduced efficiently by approximate 80% using the syngas with tar

under the test condition,and is a proper way for the resource utilization of biomass.
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Tab.1 Proximate and ultimate analysis of the wood pellet
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Fig.1 Reburning rest system for small batch size biomass gasification
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Fig.2 NO reduction efficiency changing with oxygen
flow rate at 900 'C
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Fig.5 Comparison of NO reduction efficiency
by using different fuels at 1 200 'C
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Fig.6 Comparison of NO reduction efficiency by using

syngas with tar, phenol, benzene and light gas
mixture at 1 200 'C
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